Title: Antioxidant capacities and total phenolic contents of 20 polyherbal remedies used as tonics by folk healers in Phatthalung and Songkhla provinces, Thailand

Authors:
Sasitorn Chusri (sasitorn.chu@psu.ac.th)
Nongluk Kunworarath (nongluk.ku@psu.ac.th)
Surasak Limsuwan (surasak.l@psu.ac.th)
Katesarin Maneenoon (katesarin.m@psu.ac.th)
Patcharawalai Jaisamut (patcharawalai.j@psu.ac.th)
Wipawee Chanthasri (j-miniiz_we_em@hotmail.com)
Nuntitporn Puangkeaw (toey_toey_wc@hotmail.com)
Piyapong Choochana (fankkob@hotmail.com)

Version: 3 Date: 02 Feb 2018

Dear Editor-in-Chief,

Thank you very much for your very efficient and fast review process of our manuscript BCAM-D-17-00419R2 and thank you for giving us the chance to revise our manuscript.

We have tried our best to address all the points that have been raised. I hope the responses below address your comments and answer your questions satisfactorily.

1. Please specify who identified all of the plants used in this study.

The information was added in Methods section “The plant materials were identified by a botanist, Assistant Professor Dr. Katesarin Maneenoon against reference specimens of the materia medica at the Faculty of Traditional Thai Medicine, Prince of Songkla University, Thailand”
2. Thank you for uploading the Authorship Change form, however, please can you remove this from the manuscript and email it to BMCSeriesEditorial@biomedcentral.com, in order for it to be logged in the system.

The form have send to the mentioned email.

3. Please specify where all of the cell lines were purchased from.

Even though we didn’t directly purchase Vero cell which use in our study, the link for the National Center for Genetic Engineering and Biotechnology, National Science and Technology Development Agency was added.

4. Please separate your Results and Discussion sections, this is standard for manuscripts submitted to BMC Complementary & Alternative Medicine.

The Results and Discussion sections were separated.

    Once again, thank you very much in advance for your kindness in giving us an opportunity to revise the manuscript and provide valuable information to scientists in the same field in BMC Complementary and Alternative Medicine.

Best Regards,

Asst. Prof. Dr. Sasitorn Chusri
Faculty of Traditional Thai Medicine
Prince of Songkla University
Hatyai, Thailand, 90112